Author's response to reviews

Title: The BELFRAIL (BFC80+) study: a population-based prospective cohort study of the very elderly in Belgium

Authors:

Bert Vaes (Bert.Vaes@uclouvain.be)
Agnes Pasquet (agnes.pasquet@uclouvain.be)
Pierre Wallemacq (pierre.wallemacq@uclouvain.be)
Nawel Rezzoug (reznawel@yahoo.fr)
Hassan Mekouar (mekouar_hassan@yahoo.fr)
Pierre-Alexandre Olivier (pa.olivier@gmail.com)
Delphine Legrand (delphine.legrand@uclouvain.be)
Catharina Matheï (catharina.mathei@telenet.be)
Gijs Van Pottelbergh (Gijs.VanPottelbergh@med.kuleuven.be)
Jan Degryse (jean-marie.degyse@uclouvain.be)

Version: 3 Date: 16 June 2010

Author's response to reviews:

Dear sir/miss,

Herewith you will find our revised manuscript for publication in BMC Geriatrics.

- the additional files were removed
- the title of the manuscript and table titles were adjusted
- the format of the email adress list was adjusted
- figures were removed from the manuscript and a 'Figure legends' section was added
- figures were uploaded as separate files

Sincerely,

Bert Vaes